Author's response to reviews

Title: Prioritising access to pandemic influenza vaccine: a review of the ethics literature

Authors:

Jane Williams (jane.h.williams@sydney.edu.au)
Angus Dawson (angus.dawson@sydney.edu.au)

Version: 3 Date: 23 Apr 2020

Author's response to reviews:

Have changed the text to reflect your wording - though it was obvious before. This is changed to: All data generated or analysed during this study are included in this published article and its appendix.